

Since January 2020 Elsevier has created a COVID-19 resource centre with free information in English and Mandarin on the novel coronavirus COVID-19. The COVID-19 resource centre is hosted on Elsevier Connect, the company's public news and information website.

Elsevier hereby grants permission to make all its COVID-19-related research that is available on the COVID-19 resource centre - including this research content - immediately available in PubMed Central and other publicly funded repositories, such as the WHO COVID database with rights for unrestricted research re-use and analyses in any form or by any means with acknowledgement of the original source. These permissions are granted for free by Elsevier for as long as the COVID-19 resource centre remains active. Building and Environment 94 (2015) 489-503

Contents lists available at ScienceDirect

Building and Environment

journal homepage: www.elsevier.com/locate/buildenv

From street canyon microclimate to indoor environmental quality in naturally ventilated urban buildings: Issues and possibilities for improvement

Z.T. Ai, C.M. Mak*

Department of Building Services Engineering, The Hong Kong Polytechnic University, Hong Kong

ARTICLE INFO

Article history: Received 17 September 2015 Received in revised form 16 October 2015 Accepted 17 October 2015 Available online 20 October 2015

Keywords: Natural ventilation Indoor environmental quality Buildings Street canyons Urban microclimate

ABSTRACT

Many buildings in urban areas are more or less naturally ventilated. A good understanding of the current status and issues of indoor environmental quality (IEQ) in naturally ventilated urban buildings and the association with urban microclimate is fundamental for improving their IEQ. This paper reviews past studies on (a) the microclimate in urban street canyons, (b) the potential influence of such microclimate on IEQ of nearby naturally ventilated buildings, and (c) the real-life IEQ status in these buildings. The review focuses mainly on studies conducted by on-site measurements. The microclimate in urban street canyons is characterized by low wind speed, high surface temperature difference, high pollutant concentration, and high noise level. Insufficient ventilation rates and excessive penetration of outdoor pollutants are two key risks involved in naturally ventilated urban buildings. Existing knowledge suggests that reasonable urban planning and careful building envelope design are the primary methods to ensure acceptable IEQ and maximize the utilization of natural ventilation. However, quantitative studies of both microclimate in street canyons and IEQ in buildings are still highly insufficient in many aspects, which make cross comparison and influencing factors analysis currently impossible. Based on the limitations of previous studies and the current issues of naturally ventilated urban buildings, suggestions are made for future studies to better understand and improve IEQ in naturally ventilated urban buildings. © 2015 Elsevier Ltd. All rights reserved.

Contents

1.	Intro	duction .		490
2.	Micro	oclimate	e in urban street canyons	491
	2.1.	Airflow	w	491
	2.2.	Temper	erature	493
	2.3.	Traffic	pollutants	
	2.4.	Traffic	noise	
3.	Indoo	or enviro	onmental quality in naturally ventilated urban buildings	495
	3.1.	Indoor	r ACH value	
		3.1.1.	Influence of street canyon on indoor ACH value	
		3.1.2.	Real-life ACH values in naturally ventilated urban buildings	
	3.2.	Indoor	r air quality	
		3.2.1.	Evidences of outdoor penetration	
		3.2.2.	Penetration process	497
4.	Discu	ussions ar	and recommendations	497
	4.1.	Ventila	ation rate	
		4.1.1.	Rethinking existing studies	
		4.1.2.	Integration with other systems	

* Corresponding author.

E-mail addresses: zhengtao.ai@connect.polyu.hk (Z.T. Ai), cheuk-ming.mak@polyu.edu.hk (C.M. Mak).

http://dx.doi.org/10.1016/j.buildenv.2015.10.008 0360-1323/© 2015 Elsevier Ltd. All rights reserved.







	4.2.	Indoor air quality	199
	4.3.	Indoor traffic noise	199
5.	Summ	ary and conclusions	199
	Ackno	weldgment	00
	Refere	ences	00

1. Introduction

Many buildings in a wide spectrum of geographical regions are more or less naturally ventilated, primarily because of the wellknown advantages of natural ventilation in energy saving and health effect. Driven by wind and buoyancy forces [1-3], natural ventilation has a great potential to offset energy consumption by mechanical ventilation systems [4-8]. Compared to air conditioned buildings, naturally ventilated buildings are closely associated with reduced prevalence of sick building syndrome symptoms, crosscontamination risk of airborne infectious diseases, and shortterm sick leave [9–15]. However, some buildings are naturally ventilated, essentially because their windows or doors have to be open to connect indoor with outdoor. Shops on the ground floors need open doors for business. Ordinary residential and school buildings must open windows or ventilation grilles, at least intermittently, to dilute indoor stale air, because ordinary domestic air conditioners provide no fresh air [16]. It was reported that continuous operation of such air-conditioners with closed windows and doors will rapidly elevate the indoor CO₂ concentration [17,18] and result in a deteriorated indoor air quality [8,19]. Regardless of intentions, the wide existence of natural ventilation in urban buildings highlights the need to pay special attention to their indoor environmental quality (IEQ).

IEQ in naturally ventilated buildings strongly relies on local outdoor microclimate, including particularly wind speed, air temperature, pollutant concentration and noise level [11,20-22]. Fig. 1 presents schematically the association between IEQ in naturally ventilated buildings and the microclimate in their nearby street canyon through the breath process at building envelopes. Wind speed in urban areas, especially in high-density cities, is seriously decreased [23]. Climatic data shows that wind speed at 20 m above the ground level in urban area of Hong Kong (specifically in Tseung Kwan O) has decreased from 2.5 m/s to 1.5 m/s over a 10-year period from 1994 to 2004 [24]. Low wind speeds in street canyons result in less pressure differences around buildings to drive indoor natural ventilation. It was reported that air temperature inside a street canyon is lower by 3–5 °C than the corresponding air temperature above the canyon [25] and is higher by nearly 2 °C than that in a suburban location [25,26]. In addition, traffic-related air pollution and noise are two serious environmental issues in urban street canyons. Owing to increased traffic emissions and adverse dispersion conditions including low wind speeds [27,28], pollutants can stagnate and accumulate to reach very high levels in street canyons [29–33]. Moreover, road-side noise levels measured in street canyons frequently exceed the legislated limit of 70 dBA [34–36]. Together with low wind speeds and high air temperature, such high pollutant concentrations and high noise levels could significantly deteriorate IEQ in naturally ventilated urban buildings and even hinder its use [21].

A good understanding of urban microclimate and the associated IEQ status in naturally ventilated urban buildings is fundamental for improving their IEQ. This paper provides an overview of past studies on the microclimate in urban street canyons and the associated IEQ in naturally ventilated urban buildings. To do so, it investigated relevant publications in related journals, including particularly Building and Environment and Atmospheric Environment. Major books regarding natural ventilation and urban environment were also investigated. Among these numerous publications, this paper further focused on studies conducted by on-site measurements, whereas studies by model experiments, numerical simulations, empirical and analytical models were paid little attention. On-site measurements were focused, essentially because they provide the first-hand data to reveal the real-life street canyon microclimate and IEQ in buildings, which are thus the best choice to answer the research questions defined in this paper (see next paragraph). Model experiments, such as atmospheric boundary layer wind tunnel experiments, use reducedscale models to investigate the basic structure of flow and pollutant dispersion around building(s) and provide benchmark data for numerical validations, which, however, are constrained by similarity criteria [37]. Numerical simulations, such as computational fluid dynamics (CFD) simulations, are very powerful and widely used to investigate flow related processes in the built environment, which can provide whole-field data and do not have similarity problems [38,39]. However, both model experiments and numerical simulations use simplified physical models and cannot take the influence of all environmental parameters into account. Empirical models provide a rapid but crude estimation normally for engineering applications, while analytical models increase the understanding of physical mechanisms but are limited to very simple flow problems.

Reviews in this paper is intended to answer three questions: (a) what are the urban microclimate conditions around buildings and their possible influences on IEQ in naturally ventilated buildings; (b) what are the actual indoor environmental conditions in naturally ventilated buildings in urban areas; and (c) what are the limitations of current studies and the possible areas for future



Fig. 1. Schematic view of the association between the IEQ in naturally ventilated buildings and the microclimate in their nearby street canyon; the subscripts o, s, i, v, w, l, and g of the U, T, C, L indicate outside the canyon (background), street canyon, indoor, vehicle, windward facade, leeward facade, and ground, respectively.

studies on improving IEQ in naturally ventilated urban buildings. Followed by this introduction, Section 2 reviews microclimate in urban street canyons. Section 3 reviews the IEQ in naturally ventilated urban buildings including particularly the influence of urban microclimate. Section 4 discusses existing studies and makes recommendations for future studies. Section 5 concludes this review work.

2. Microclimate in urban street canyons

This section provides an overview of the microclimate in urban street canyons, including airflow, temperature, pollutants and noise. Understanding urban microclimate is the basic prerequisite of understanding and thus improving IEQ in naturally ventilated urban buildings.

Air quality in skimming-flow deep canyons [23] was mostly investigated, as they represent the worst condition for pollutants dilution [91].

2.1. Airflow

Compared to rural areas, wind speed in urban areas is seriously decreased as a result of increased roughness caused by various constructions and complex streets [23–25]. Decrease of wind speed in street canyons poses an important limitation to apply natural ventilation in dense urban areas [40].

Flow pattern in an urban area depends on the building disposition, particularly the aspect ratio AR (ratio of the mean building height H to the street width W) [23]. When the buildings are widely spaced (AR < 0.3–0.4), their flow fields do not interact and thus are almost the same with those around isolated buildings. At closer spacing (AR up to 0.65–0.7), the wake of an upstream building is interfered by the flow field of its downstream building. At even greater spacing (AR > 0.65-0.7), the ambient flow skims over the building tops; such a flow regime is called skimming flow. The study of wind environment in skimming-flow street canyons were paid special attentions, as they are generally considered to be with a worse wind condition compared to those in lower AR streets. Flow patterns inside a street canyon are very distinctive in response to different ambient wind conditions above the canyon [11,23,25,41–45], which are particularly associated with the along and the across canyon velocity components above the canyon. Under a parallel ambient flow, if the wind speed above the canyon is high, a strong relationship between the outside and the inside flows can be found. Wind inside the canyon flows along the canyon axis with possible uplifts along the building walls, due to the friction of building walls and ground surface. Under a perpendicular ambient flow, if the wind speed above the canvon is high, the canyon flow can be seen as a secondary circulatory flow driven by the imposed flow above the canyon. Under an oblique ambient flow, which occurs in most time, helical vortexes could be developed along street canyons. For all ambient wind directions, if the wind speed above the canyon is low, the coupling between the upper flow and the canyon flow is lost, where thermal flows and mechanical disturbances play a significant role in shaping the wind flow patterns inside the canyon.

The threshold ambient wind speed depends on the canyon AR. According to [46], who worked with a street canyon with AR equal to 1.4, the threshold value to establish the coupling and to form circulatory vortexes inside the canyon is between 1.5 m/s and 2.0 m/s. Similar values are reported by Refs. [41,42,47,48]. They worked with canyons having aspect ratios ranging from 1 to 1.52. However, as shown by Ref. [49], for canyons with a higher AR (that is 2.5 in their study), the coupling is established under a much higher ambient wind speed ranging approximately between 4 m/s and 5 m/s (corresponding to the across canyon wind speed equal to 2-3 m/s). It was also stated that the correlation between the wind speed inside and above the canvon is unclear when the ambient wind speed was below this range [49]. The value 4 m/s was defined as the threshold wind speed in their later studies [50,51] of predicting wind speeds inside street canyons. Comparison of these results may imply that the threshold wind speed increases with the increase of AR value. This should be explained by that a faster wind is required to penetrate into a deeper street canyon. However, the influence of other factors, such as meteorological condition and buoyancy force, on this threshold cannot be revealed from these studies [41,42,46–51].

Some studies measured simultaneously wind speeds inside and outside (mostly above) a canyon, which are summarized in Table 1. Note that there could be several measurement locations along

Table 1

Comparison of measured wind speeds inside and above canyons; climate classification is made according to Köppen–Geiger Climate Classification System [52]; U_o is ambient wind speed, U_{o-m} mean ambient wind speed, $U_{o-p\%} p\%$ of ambient wind speeds, and similar naming method for wind speed inside canyon U_i , L/H ratio of street length L to building height H.

City/climate	Measurement period	AR	L/H	Ambient wind speed (m/s)	Ambient wind direction	Wind speed inside canyon (m/s)	Reference
Stafford/Maritime temperate (Cfb)	Mar.—Dec. 1997	0.59	_	$U_{o-m} = 6.25U_{i-m}$	-	Total: $U_{i-m} = 0.16U_{o-m}$	[53]
Kyoto/Warm temperate (Cfa)	-	1	_	$U_{o-m} = 1.49U_{i-m}$	-	Total: $U_{i-m} = 0.67U_{o-m}$	[41]
Watford/Maritime temperate (Cfb)	-	-	-	$U_{\text{o}-m} = 1.59U_{i-m}$	-	Total: $U_{i-m} = 0.63U_{o-m}$	[54]
Athens/Dry-summer (Csa)	Jul. 19–23, 2002	1.7	2.25	$U_{o-m} = 3.4$ $U_{o-25\%} < 2.1$ $U_{o-95\%} < 6.6$	Perpendicular: 21% Parallel: 13% Oblique: 66%	Horizontal: $U_{i-50\%} < 0.6$, $U_{i-95\%} < 1.5$ Vertical: $U_{i-m} = 0.34 - 0.46$	[55]
Goteborg/Maritime temperate (Cfb)	Jun. 2003–Aug. 2004	2.1	3.3	$U_{o-m} = 1.8 - 2.5$	-	Total: $U_{i-m} \leq 0.5$	[45]
Athens/Dry-summer (Csa)	Jul. 7—15, 1997	2.5	7.1	$U_{o-93\%} < 5.0$ $U_{o-50\%} < 2.0$	Perpendicular: 7% Parallel: 12% Oblique: 81%	Total: $U_{i-45\%} < 0.3$, $U_{i-96\%} < 1.0$ Horizontal: $U_{i-m} = 0.35$ Vertical: $U_{i-m} = 0.26$	[49]
Ghardaia/Arid (BWh)	Feb. 19–24, Sep. 6–13, 2005; Aug. 6–15, 2006	2.7-2.9	9.4-11.6	$U_{o-m} = 1.0 - 3.0$	_	Total: $U_{i-m} = 0.5 - 1.5$	[56]
Athens/Dry-summer (Csa)	Sep. 2001	3.3	1.7	$U_{o-(89\%-94\%)} < 4.0$	Perpendicular:28% Parallel: 16% Oblique: 56%	Total: $U_{i-m} = 1.0 - 2.0$	[25]
Tinos/Dry-summer (Csa)	Summer and winter	0.7–0.9 2.0–4.0	-	$U_0 = 9.0 - 16.0$	-	$U_{i-m} = 1.0 - 6.5$ $U_i < 2.0, U_{i-m} \approx 1.0$	[26]



Fig. 2. Relationship between wind speeds inside and above a canyon with AR equal to 2.5 [49]: (a) variation of the total wind speed inside the canyon (U_{i-t}) as a function of the along canyon wind speed above the canyon (U_{o-x}) , and (b) box plot of the along canyon wind speed inside the canyon (U_{i-x}) against various classes of the parallel to the canyon wind speeds above the canyon.

vertical and/or horizontal directions inside a canyon and their statistical values are presented. Comparison of these wind speeds inside and outside street canyons leads to four remarks.

First, regardless of other influencing factors, a significant decrease of wind speed inside a street canyon is evident in comparison with wind speed outside the canyon. Fig. 2 presents the statistical relationship between wind speeds inside and above a street canyon with an AR equal to 2.5. The ratio of wind speed inside a canyon to that outside the canyon is mostly in the order of 10–30%, whereas relatively large ratios (over 0.6) are reported by Refs. [41,54]. Variation in this ratio should be attributed to many influencing factors, such as topography, gross building coverage ratio [57], canyon AR [25,40,44], locations of measurement, and ambient wind speed and direction above the canyon. In addition, if the coupling of wind flow between canyon interior and its above atmosphere is not established, this wind ratio should be of little significance. Even if the coupling is achieved, the relationship between the wind ratio and other influencing factors is still unknown.

Second, wind direction inside street canyons is complex, which is dominated by the along canyon flow combined with upward and downward movements [49,55]. Fig. 3 presents the hourly averaged wind flow patterns on an across canyon vertical plane, where the upward and downward flows are strong and the normal-to-facade flows are very weak. In addition, it was reported that mean wind speeds in the canyon centerplane generally increase with height [25]. These first two remarks indicate the importance of considering urban effects in natural ventilation studies.

Third, wind speed inside a canyon does not necessarily decrease with the increase of canyon AR. Apart from AR, there are other important factors governing the flow field inside a canyon. For examples, it was found that thermal effects [41], end effects, or finite length canyon effects [48], canyon length ratio (L/H) [43], and traffic induced turbulent effects [43] play important roles in shaping the flow characteristics inside a canyon.

Fourth, measurements are almost independent from each other in terms of meteorological condition, measurement location,



Fig. 3. Hourly averaged wind flow patterns inside a canyon with AR equal to 2.1 from 6:00 a.m. to 17:00 p.m. on a specific day, with the mean ambient wind speed of 1.8 m/s, ranging from 0.2 m/s to 4.7 m/s [45]. The numbers in the upper left indicate the start hour in a day. Black vectors are observed fields in the across canyon plane (a reference vector is shown in the 6 h plot). The contours show the interpolated along canyon wind speed within the canyon. The vector emanating from the circle at top center is the reference wind on the along canyon plane, where the right direction represents the east.

measurement duration and data presentation, which makes cross comparison between different measurements impossible. Particularly, the influence of climate characteristics cannot be revealed from short-term measurements, as the meteorological condition during measurements and the measurement period in a year have major influences on the measured results. Therefore, long-term measurements, such as one year or at least one season, are useful to reveal the general wind conditions for a specific urban configuration in a specific climatic region.

2.2. Temperature

Air temperature inside a street canyon strongly determine the cooling potential of natural ventilation of its nearby buildings [25], while temperature difference between canyon surfaces is an important driving force of flow movements in a canyon particularly when the ambient wind flow is slow.

Surface temperatures of a canyon are closely associated with local climate, urban morphology, thermal properties of building and street materials, canyon orientation, canyon AR, and sky view factor. Fig. 4 presents a schematic view of the thermal environment inside a street canyon, where the maximum temperatures and temperature differences reported in two previous studies [49,58] are also included. In general, due to different solar incidence angle, horizontal ground surfaces have much higher temperatures than vertical wall surfaces [58-62]. With a same solar incidence angle, a surface with a lower albedo can reach a much higher surface temperature than that with a higher albedo [58]. The change of solar incidence angle with time could lead to a significant change in surface temperatures inside a canyon, with daily amplitudes up to 35 °C, while the maximum temperature between two opposite building walls with and without direct solar radiation was observed to be up to 19 °C [49]. In addition, a canyon with a higher AR value has a more restricted sky view factor and thus can produce a stronger shading effect to solar radiation [26,60,63,64] than a canyon with a lower AR value. This supports considering high AR urban configurations in low latitude geographical regions, where minimization of solar radiation is a desirable criterion in urban design [60.64]. Compared to the results measured in a canyon with AR equal to 2.5 [49], relatively lower surface temperatures and temperature differences were reported in a deeper canyon with AR equal to 3.3 [25], where the maximum surface temperature during the daytime varies between 35 °C and 41 °C and the maximum daily simultaneous temperature difference between the two opposite facades, as a function of the canyon's height, was between 9 °C and 18 °C.

Air temperature inside a street canyon is much lower (up to 13 °C) than the temperature on canyon surfaces of direct solar radiation [49]. However, in contrast to surface temperature, canyon configurations and surface materials do not cause an obvious influence in air temperature [26], provided that the AR value is not extremely high (e.g., higher than 7–10 in Ref. [65]). Slightly lower air temperatures (by 1–3 °C) were observed in narrower street canvons with AR ranging from 2 to 4 than those in wider street canyons with AR ranging from 0.7 to 0.9 during the daytimes, while such temperature differences were 3–4 °C during the nighttimes [26]. It is thus believed that the average air temperature is determined dominantly by large-scale regional factors rather than street-scale factors [59,66,67]. Vertically, no pronounced temperature stratification as a function of canyon's height was observed, which should be attributed to the strong mixing and advective phenomena inside the canyon [49]. However, due to shading effect, the air temperature inside a deep canyon with the AR equal to 3.3 during the daytime was observed to be 3-5 °C lower than the corresponding air temperature above the canyon [25]. In addition, the heat island effect, namely the air temperature difference between a monitoring location inside the canyon and a background station, was mostly close to 2 °C [25,26]. However, the air temperature difference between the monitoring location above the canyon and the background one was between 4 °C and 11 °C [68,69].

The above review provides a basic understanding of thermal environment in street canyons, although cross comparison of different measurements and influencing factors analysis are difficult. From above review, two remarks can be made. First, under low latitude geographical regions, deep street canyons with high ARs are an important design criterion in urban development in order to reduce solar radiation into the canyons. However, as reviewed in the previous Section 2.1, deep canyons normally correspond to stagnant wind speeds, which lead to negative effects on removing pollutants. In this context, exploring optimal AR values are still needed when taking into account solar gain, city ventilation and pollutant removal. Second, anthropogenic heat and use of low albedo (and high absorbing) street materials are important reasons causing higher temperatures inside street canyons [49]. Akbari et al. [70] reported that changing the surface albedo from 0.25 to 0.4 in a typical mid-latitude warm climate can lower afternoon air temperatures on summer days by 4 °C. Owing to the heat rejection by air-conditioners, the recessed spaces outside high-rise residential buildings in Hong Kong have a maximum temperature elevation of 11 °C [71].



Fig. 4. A general view of thermal environment inside a street canyon, where the major causes of temperature differences, and the maximum temperatures and temperature differences reported in two references [49,58] are provided; symbols T_a , T_{g-gp2} , and T_{g-as} indicate temperature of air, temperature of sunlit ground with gray pavement, temperature of shaded ground with gray pavement, and temperature of sunlit ground with asphalt street, respectively.



Fig. 5. Major influencing factors of pollutants level and their distribution inside a street canyon: traffic intensity, mass exchange rate between street canyon and its upper atmosphere, and wind flow pattern.

2.3. Traffic pollutants

Traffic exhausts have long been recognized as one of the major anthropogenic sources of air pollution in urban areas [72–76]. The main traffic-related pollutants are PMs, CO, NOx, hydrocarbons [77,78]. Due to the increased traffic emissions and/or adverse dispersion conditions [27], pollutants can accumulate to reach very high levels in street canyons in comparison with background concentrations [29–33,79–86]. Traffic-related pollutants like PM₁₀, CO, benzene and toluene in street canyons were reported to frequently exceed the safety thresholds suggested by local or WHO air quality standards [87,88]. High levels of traffic-related pollutants can seriously deteriorate urban air quality and cause damage to human health [89,90].

Similar with flow pattern, air quality in skimming-flow deep canyons [23] was mostly investigated, as they represent the worst condition for pollutants dilution [91]. Under a specific environmental condition, a narrower canyon generally corresponds to a worse street-level air quality [92]. Dispersion and dilution of airborne pollutants in a street canyon rely strongly on the exchange rate of flow between a canyon interior and its upper atmosphere [93,94]. When the ambient wind speed above a canyon is too low to establish the coupling between the canyon interior and its upper atmosphere (see Section 2.1), pollutants accumulate in the street canyon [46,95–97]. On the contrary, when the ambient wind speed is sufficiently high to establish such a coupling, pollutants can be effectively diluted to the upper atmosphere. The pollutant concentrations and pedestrian exposure levels in street canyons are closely associated with the traffic intensity [98–101]. Typically, high concentrations of traffic-related pollutants, such as PM_{2.5}, PM_{10-2.5}, black carbon (BC) and monocyclic aromatic hydrocarbons, in street canyons were reported in rush hours on workdays [102–104].

Fig. 5 presents schematically a spatial concentration distribution inside a street canyon. In general, horizontally, traffic-related pollutants maintain relatively high concentrations in a region near the leeward side of a canyon [95,100,105,106], while concentrations decrease vertically along height above the ground on both sides of the canyon [27]. Vakeva et al. [107] reported that the concentrations of submicron particles and gaseous pollutants emitted at street level decrease by 80% at the height of 25 m. Concentration gradients in vertical profiles of fine particles are still obvious [27]; it is 12% between two measurement levels at 2.5 m and 8.9 m. They explained that the decrease of canyon concentrations in the upper part of the canyon is due to the enhanced turbulence and mixing. In contrast to fine particles and gases, due to the gravitational effect, significant vertical concentration gradients were reported for larger-sized particles [108].

Table 2 summarizes the reported ratios of pollutant concentration measured in a street canyon to that in a corresponding background location, such as urban park and suburban area. These concentration ratios are mostly between 1.0 and 3.0, indicating high contrasts of pollutant concentration between traffic streets and background locations. However, dependent on traffic intensity, the concentration ratios should vary in both space and time. In addition, owing to the limited number of on-site measurement, the influence of climate characteristics and the measurement period cannot be revealed from the results summarized in Table 2. Largescale and long-term measurements would be useful for cross comparison and influencing factors analysis.

2.4. Traffic noise

Noise is another serious environmental issue in urban areas. Although a combination of many environmental noise sources contributes to shaping an urban soundscape, road traffic noise is usually the most important contributor. Exposure to excessive noise levels could result in a series of negative health effects [112], particularly hypertension [113–115].

Noise levels in street canyons are strongly associated with traffic intensity [36]. Previous on-site measurements (see Table 3) indicate that road-side noise levels in street canyons frequently exceed the legislated limit of 70 dBA [34]. Note that one study [116] reports median noise levels, while others report mean values. Exception are those reported by Morelli et al. [116], who measured noise levels in three European cities, which are all below though still close to 70 dBA. Owing to the uneven distribution of traffic intensity, these reported traffic noise levels in specific street canyon(s) could not represent the average noise level in a city, which, however, are sufficient to demonstrate the severe sound environment in urban areas. Such high levels of traffic noise would certainly lead to a deteriorated indoor sound environment.

Table 2

Ratios of pollutant concentration measured in a street canyon to that in a corresponding background location; climate classification is made according to Köppen–Geiger Climate Classification System [52]; PAHs is the abbreviation of polycyclic aromatic hydrocarbons, while Cr, Cu and Fe for chromium, copper and iron, respectively.

City/climate	Measurement period	PM ₁₀	PM _{10-2.5}	PM _{2.5}	BC	Cr, Cu and Fe	NOx	NO_2	NO	CO	VOCs	PAHs	Benzene	Reference
Guangzhou/Warm temperate (Cwa)	Feb. 1–2, 2002	1.2	_	_	_	_	-	_	_	4.7	_	_	1.4	[86]
Hong Kong/Warm temperate (Cwa)	Jan. 1 – Dec. 8, 2005	_	1.3	1.5	_	-	_	_	_	_	-	_	-	[102]
5 Dutch cities/Maritime temperate (Cfb)	6-month beginning in	1.2	-	1.2	1.9	2.0-3.0	1.8	1.5	_	_	-	_	-	[31,109]
	Jun. 2008													
Copenhagen/Maritime temperate (Cfb)	JanMar. 1992-1994	_	-	_	2.4	-	_	_	-	_	-	3.2-5.3	-	[110]
Raleigh/Warm temperate (Cfa)	Aug. 11 - Sep. 20, 1988	_	-	_	_	_	_	_	-	_	1.7	-	-	[111]
Los Angeles/Dry-summer (Csa)	Feb. 14, 20, Apr. 7, 16,	_	-	_	_	_	_	2.0	13.9	2.6	_	-	-	[99]
	24, 2003													

Table 3	
Traffic noise levels measured	in urban areas.

City	Description	Mean noise level (dBA)	Variation (dBA)	Reference
New York	329 samples at 99 street sites	73.4	55.8-95.0	[36]
New York	6-day measurements near a major urban highway	74.5-76.9	Standard deviation: 1.4–1.6	[35]
Valencia	5-year measurements at a square	72.3-73.4	69-77	[119]
Curitiba	100 samples at several locations on two traffic streets	73.1	68-78	[120]
Athens	Short-term measurements at 9 street canyons	75	66.1-81	[117]
Basel	20-min measurements at 60 street sites	62*	55-66	[116]
Girona	20-min measurements at 40 street sites	64*	60-68	
Grenoble	20-min measurements at 41 street sites	67*	62-69	

Note: * indicates median value.

Nicol and Wilson [117] measured the vertical variations of noise level in nine street canyons with AR varying from 1.0 to 5.0. Based on these measured data, they then developed a mathematical model to predict noise levels inside a street canyon as a function of street width and height above the street (see Fig. 6), which are intended to provide guidances for building design. In general, noise level increases with the increase of street width, which is attributed to that a wider street corresponds to a higher traffic intensity. Moreover, noise level attenuates along canyon height. Therefore, noise level is easier to exceed 70 dBA at the lower part of street canyons. In a deep canyon in Hong Kong [118], the noise level at several lower floors (1/F-3/F, 5/F) is greater than 70 dBA.

3. Indoor environmental quality in naturally ventilated urban buildings

Microclimate in street canyons would certainly influence the IEQ in nearby buildings (as indicated in Fig. 1). Only air exchange rate (ACH) and indoor air quality (IAQ) are included in this section, whereas indoor temperature and noise level are not. The influence of outdoor temperature and noise on indoor environment is almost instant and easy to be sensed by occupants. When suffering from intolerable temperature and noise level, active control measures, such as closing windows and turning on air conditioners, can be taken immediately. As a result, studies regarding indoor temperature and noise level were paid mostly to exploring subjective sensation and comfortable range, which are out of the scope of this paper. On the contrary, insufficient ACH values and excessive indoor air pollutants are difficult to detect by human body, which thus could cause serious negative health effects to occupants. Compared to temperature and noise level, ACH values and IAQ in naturally ventilated urban buildings have received much more attentions in past decades.



Fig. 6. Predicted noise levels (dBA) as a function of street width and height above the street [117]. The region (OK) with noise level lower than 70 dBA is acceptable for natural ventilation design, whereas the region (Not OK) with noise level higher than 74 dBA is unacceptable. There are possibilities for design solutions in the region between 70 and 74 dBA.

3.1. Indoor ACH value

3.1.1. Influence of street canyon on indoor ACH value

It has been confirmed in Section 2.1 that wind speed inside a street canyon is significantly decreased compared to that above the canyon. Decreased wind speeds would result in lowered pressure differences around buildings to drive indoor natural ventilation. Therefore, natural ventilation performance in urban buildings drops largely compared to rural areas [121]. van Hooff and Blocken [122] evaluated quantitatively the influence of surrounding buildings on natural ventilation performance of a large semi-enclosed stadium using CFD method. It was found that neglecting the surroundings can lead to overestimations of the ACH up to 96% (see Fig. 7). Gao and Lee [123] worked with a group of high-rise residential buildings and found that the presence of surrounding buildings lowers the wind speed in the vicinity of buildings by 2.5%–86.8%. In addition, Georgakis and Santamouris [25] compared ACH values calculated using climate data measured inside and above a canyon with AR value equal to 3.3. Results show that, inside the street canyon, ACH values for single-sided and cross ventilation are reduced by 82% and 68%, respectively. These studies indicated quantitatively the significant blocking effect of surrounding buildings and thus suggested explicitly the importance of describing the surroundings in detail when assessing natural ventilation performance in buildings [124]. Among influencing factors, a better building disposition, particularly a wider street, is very important to enhance air penetration and movement inside street canyons and thus potentially improve natural ventilation performance in buildings [125,126].

3.1.2. Real-life ACH values in naturally ventilated urban buildings

On-site measurements of ACH values in naturally ventilated urban buildings are valuable to reveal the real-life ventilation condition in such buildings (see Table 4). Analysis of these real-life



Fig. 7. Comparison of the predicted ACH/ U_{10} for eight wind directions between cases with and without surrounding buildings [122], where U_{10} is reference wind speed at the height of 10 m and φ is wind angle. At the wind angle of 196°, the overprediction of ACH/ U_{10} is 96% when the surrounding buildings are not included.

Building type	Location	AR	Mean ambient wind speed	Indoor/outdoor temperature difference	Location on a building	Opening area	Measurement technique	ACH-SS ^a (h^{-1})	ACH-cross ^b (h ⁻¹)	Reference
Residential	Athens	1.4	1	1	3rd/5	I	Tracer gas decay $(N_2O \text{ and } SF_6)$	2.6-3.1	10.2-15.4	[127,128]
buildings		2.6	1	1	2nd/5	I		3.9 - 5.8	9.8 - 10.1	
	Athens	3.3	Parallel to the canyon 3.2–3.3 m/s	0.05 °C	3rd/5	1.4 and 1.6 m ²	Tracer gas decay (N ₂ O)	0.4 ± 0.1	1.8 ± 0.3	[25]
	Hong Kong	irregular	1.9–4.6 m/s	<1.0 °C	12th/27 10th/22	1.2–3.5 m ²	Tracer gas decay (CO ₂)	1.5 - 16.1	Ι	[129]
	Beijing	I	<0.5 m/s outside windows	<6.0 °C in summertime	Several rooms in a six-story building	0.4 m ²	Constant tracer gas injection (CO ₂)	<1.6	1	[17]
Hospital buildings	Bradford	I	1.0-4.0 m/s	4.6 °C	3rd/3	0.1 m ^c	Tracer gas pulse injection method (CO ₂)	I	3.4-6.5	[130]
)	Hong Kong	p	1.3-4.8 m/s	<1.0 °C	I	I	Tracer gas decay method (SF ₆)	14 - 31.6	11.9-69	[121]
	Lima	70 rooms in 8 hospitals	I	I	I	I	Tracer gas decay method (CO ₂)	I	18-46 ^e	[13]
Notes: ^a single-si another is surrou	ded natural ver inded by high b	ntilation; ^b cross uildings; ^e interu	natural ventilation; ^c wind quartile range.	lows had a locking r	nechanism that re	stricted the maxim	um opening to 0.1 m; $^{\rm d}$ one is located	on a hillside surro	unded by green envir	onments and

Table 4

ACH data leads to three important observations.

First, the reported ACH values vary significantly between different measurements and some of them are insufficient (or intermittently insufficient) for maintaining acceptable IAQ, which indicates basically that natural ventilation performance is determined by many factors, such as wind speed and direction, opening configuration, surrounding characteristics, and floor location in a building. This is a well-known disadvantage of natural ventilation technique. Georgakis and Santamouris [25] observed very low ventilation rates, probably because the outdoor wind was nearly parallel to the canyon axis during the measurements. Relatively small ventilation rates obtained by Gilkeson et al. [130] could be attributed to the fact that the maximum opening was 0.1 m due to a locking mechanism. The one-year measurement in four-students dormitories by Li et al. [17] showed that single-sided natural ventilation could not provide sufficient outdoor air to dilute the indoor CO₂ in 18% of the measured period, while ACH values were all less than the recommendation $(2.3 h^{-1})$ of IAQ standards [131]. In addition to small window area $(0.4 m^2)$, insufficient ventilation rates should be attributed to that during approximately 73% of the measurement period the outdoor wind speed was slower than 0.5 m/s. With similar reasons, Santamouris et al. [132] reported that insufficient ventilation rates occurred during teaching periods in approximately 77% of the measured classrooms and about 52% of the classrooms presented a mean indoor CO₂ concentration higher than 1000 ppm. In addition, influence of wind speed and floor location in a building on natural ventilation performance is very significant [133–139], although it is not explicitly revealed in these measurements (listed in Table 4).

Second, despite of being unstable, natural ventilation could potentially provide very high ventilation rates, particularly when cross ventilation is achieved. Hospitals especially benefit from this advantage of natural ventilation, as they normally require a high ventilation rate, namely 6 h⁻¹ for general ward environments and 12 h⁻¹ for isolation wards [140,141]. Regarding the unstable characteristics, hybrid ventilation combining natural and mechanical ventilations should be a promising option, which is discussed in Section 4.

Third, in addition to wind effect, buoyancy effect due to temperature difference is another driving force of natural ventilation. However, these measurements (listed in Table 4) indicated that indoor/outdoor temperature differences in naturally ventilated buildings, particularly in residential buildings, are very small, less than 6.0 °C even in the cold climatic region, Beijing [17]. Although the indoor/outdoor temperature differences can be much larger in wintertime [17], natural ventilation would not be (fully) used in such conditions. With such small temperature differences, windinduced pressure differences are mostly the dominated driving force for natural ventilation.

3.2. Indoor air quality

3.2.1. Evidences of outdoor penetration

A large amount of literature reported indoor/outdoor (I/O) concentration ratio of PMs and other pollutants, which are intended to reveal the relative relationship of pollutant concentration between the indoor and the outdoor environment. However, I/O ratios of general pollutants that can be generated from both the indoor and the outdoor environment are hardly helpful for understanding the indoor and outdoor concentration relationship, as they vary considerably among different studies [142–144]. Such variations could be attributed to that the I/O ratios depend strongly on indoor and outdoor emission rates, window or crack characteristics and air exchange rates. Studies of PM levels inside and outside naturally ventilated residential buildings [145], commercial

office buildings [146] and school buildings [147] indicated that indoor activities are the main contributor of indoor PM levels.

However, there are sufficient evidences demonstrating that outdoor pollutants can easily penetrate into its nearby indoor environment through flow exchanges [146,148-151]. Fischer et al. [152] reported that outdoor PM₁₀ and PM_{2.5} concentrations are 15–20% higher at homes located in high traffic intensity streets in Amsterdam than those in low traffic streets, while outdoor concentrations of the particulate components Benzoapyrene (BaP) and total PAHs, as well as the gaseous components benzene and total VOCs are almost 100% higher at high traffic intensity homes. Similarly, I/O ratios for PM_{2.5} were found to be higher at the trafficked site than the residential site [153]. A comparative study by Lai et al. [154] found that the I/O ratio of ozone for using natural ventilation is significantly higher than for using mechanical ventilation with filtrations. Moreover, Rivas et al. [155] reported that the indoor levels of traffic sourced pollutants, such as NO₂, equivalent black carbon (EBC), ultrafine particles (UFP) and stibium (Sb), are very similar to those recorded outdoors.

Although many studies confirmed the contribution of outdoor pollutants into the indoor air pollution, only a few studies clarified quantitatively such a contribution. Particularly, Zhu et al. [156] measured UFP in and out of four buildings near a freeway, where there were no indoor UFP generation sources. They found that I/O ratios under natural ventilation with open windows are close to 1.0 (see Fig. 8), which are substantially higher than those under infiltration and mechanical ventilation. Kliucininkas et al. [157] measured indoor and outdoor PAHs in six urban homes. When there is no indoor PAHs source, the I/O ratio ranges from 0.05 to 0.36, which can be regarded as the penetrated portion from the outdoor environment. Amato et al. [158] analyzed the indoor and outdoor sources of PM2.5 in 39 primary schools. They found that approximately 47% of indoor PM_{2.5} is generated indoors and 53% outdoors. Among outdoor sources, the traffic contribution was highlighted, suggesting avoiding orienting windows directly to the traffic street. Meier et al. [159] modeled indoor air pollution originated from the outdoor environment using regression models, revealing that the fractions of outdoor pollutants entering indoors are between 30% and 66% except for the coarse particles. These quantitative studies indicate explicitly that outdoor air pollutants are an important contributor of indoor air pollution. However, the amount of penetration is still highly case-dependent.

3.2.2. Penetration process

The studies reviewed in Section 3.2.1 measured I/O ratios in practical conditions, where the pollutant sources were uncontrollable and the emission rates were uncertain. A few studies

investigated the dynamic process of outdoor pollutants penetrating into the indoor environment under a controlled outdoor pollutant source. Santos et al. [160] experimentally measured the pollutant concentration and its fluctuation in and around a one-story, tworoom, compartment building placed on an outdoor open space, with a pollutant source located in front of the building model. The highest concentration and concentration fluctuation intensity were found near the windward walls, while the indoor concentration levels responded quickly to outdoor source and were much higher under naturally ventilated condition than under mechanically ventilated condition. Yang et al. [161] simulated the dispersion of outdoor traffic pollutants into the nearby buildings, focusing on examining the influence of opening ratio (ratio of opening area on a facade to the facade area) of a building on pollutant penetration. Results show that the amount of traffic pollutants entering into buildings through unit ventilation area decreases as the increase of opening ratio.

Apart from pollutants generated outside of buildings, recent studies highlighted that pollutants, particularly toxic gases or infectious biological aerosols, generated inside a room can reenter its nearby rooms in a same building through envelope vents. Such kind of pollutant dispersion is different from purely indoor pollutant dispersion, as the pollutant source is not located in the affected room. It was reported that, driven by wind pressure, CO produced from a house-used natural gas water heater installed in the balcony could diffuse into the adjacent room through crack infiltration [162], where the indoor CO concentration levels could reach a pronounced level under adverse ventilation condition. Motivated by the transmission of Severe Acute Respiratory Syndromes (SARS) within a same building, a series of studies [135-137] were conducted to investigate the pollutant dispersion from one unit to its adjacent unit in same multistory buildings through open windows via airflow route. The dispersion routes and reentry possibilities of such an inter-unit transmission in five-story buildings were revealed quantitatively (see Fig. 9 for examples), which confirmed its validity and severity. In general, these emerging modes of pollutant penetration within a same building pose a serious threaten to IAQ, which should be paid more attentions in the future.

4. Discussions and recommendations

Many studies have revealed the significant influence of microclimate in street canyons on IEQ in naturally ventilated buildings in urban areas, which mainly includes decreased ventilation rate and excessive penetration of pollutants and noise. Energy, daylighting, thermal comfort, human adaptation, security and privacy issues involved in naturally ventilated buildings are also very important,



Fig. 8. Measurements of indoor and outdoor UFP in buildings within 60 m from the center of a freeway in Los Angeles, CA when there were no indoor UFP generation sources [156]: (a) locations of sampling apartments (Apt. 1–4) and prevailing wind directions; (b) statistical I/O ratio as a function of particle size under natural ventilation condition.



Fig. 9. Mean reentry ratios and dispersion routes in two generic five-story buildings without and with balconies under a normal incident wind condition [136]: left two for windward side and right two for leeward side. Reentry ratio is defined as the fraction of pollutant that is generated from an index unit and reenters another unit.

which, however, are not included in this study. Based on the review of microclimate in street canyons and IEQ in naturally ventilated buildings, this section discusses implications and limitations of existing studies and then makes recommendations for future studies.

4.1. Ventilation rate

4.1.1. Rethinking existing studies

Existing on-site measurements (see Section 2.1) of airflow in urban street canyons have indicated that wind speeds inside street canyons are relatively small and wind directions are mostly parallel (or closely parallel) to building facades, which are very different from those found in rural areas or open spaces. Existing studies also suggested that building disposition (urban planning) based on local environmental conditions strongly determines the microclimate in street canyons [125,126]. The reason is that a well-planned urban area corresponds to an effective flow exchange between the canyon interior and its above atmosphere, which will not only ensure a moving flow in street canyons but also dilute polluted air. The former is the essential element to produce pressure differences to drive natural ventilation in buildings, and the latter helps to create a good outdoor environment for natural ventilation. Influenced by characteristics, meteorological climate condition, urban morphology, building disposition, and building height, the microclimate, particularly the wind condition, in a street canyon are very complex and are far from well understood from existing studies. In addition, the influence of these factors on the measured results cannot be revealed from existing studies. Particularly, short-term measurements are not useful to reveal the influence of climate characteristics. It is suggested that large-scale, long-term, highresolution, standardized on-site measurements in typical street canyons should be conducted to accumulate detailed local wind data. Representative measurement campaigns in the past include the European projects URBVENT [11] and RESHYVENT [163], as well as the French project PRIMEQUAL [22]. These data can be (a) summarized and analyzed to find their correlations with other influencing factors, such as AR value, prevailing wind speed and direction and then to optimize urban planning to improve the micro-environment in street canyons, (b) analyzed to directly correlate with natural ventilation performance of nearby buildings, and (c) generalized to provide accurate boundary conditions for further CFD analysis of optimal envelope design to adapt urban environment.

Existing studies (see Section 3.1) have confirmed the significant decrease of ACH values in urban context and have provided examples of the real-life ACH values in naturally ventilated urban buildings. They imply that careful envelope designs can increase the adaptation of buildings to the urban environment and improve natural ventilation performance in terms of increasing ventilation rate and extending its operational time. However, due to the uniqueness of envelope and surrounding characteristics of a specific building in urban areas and the inconstant wind condition, these previous studies are very hard to provide a general conclusion that is applicable to other scenarios. This is the essential reason why (a) different previous studies (Section 3.1.1) reported different percentage decrease of ventilation rate caused by the presence of surrounding buildings and (b) some studies reported well acceptable ACH values but others insufficient (Section 3.1.2). In addition, the influence of environmental parameters, such as wind direction [164,165], on indoor ACH in urban buildings are still unclear. Such knowledge is essential to guiding urban planning and building design. In order to improve our understanding of and provide general information for naturally ventilated building design in urban areas, future studies may pay more attentions on two aspects. First, more on-site measurements of ACH value in naturally ventilated urban buildings should be conducted. Long-term and largescale measurements are very useful to obtain general knowledge of natural ventilation performance in urban buildings and to analyze influencing factors. Second, future CFD studies should include more realistic boundary conditions. However, CFD studies of naturally ventilated buildings in complex urban context is still a challenge to current computational power, as the large difference in scale between windows and urban area will involve enormous amount of grid cells.

4.1.2. Integration with other systems

Careful urban planning and building envelope design are primary ways to increase natural ventilation performance of buildings in urban areas. However, they alone are limited and insufficient ventilation rates occur sometimes. In view of increasingly stagnant wind flow in high density urban areas, it is important to explore other passive and/or active systems to assist natural ventilation systems to maximize its use.

Hybrid ventilation [163,166–169] that combines natural ventilation system with either mechanical ventilation or even air conditioning systems can ensure a minimum ventilation rate when the purely natural ventilation system cannot provide sufficient airflow exchanges. A more detailed discussion of hybrid ventilation is provided in Section 4.2. In addition, architectural features, such as windcatcher [170,171], balcony [133,134,172], wing wall [173,174] and ventilation shafts [175], are promoted to use in buildings to enhance indoor and outdoor airflow exchange via ingeniously utilizing the aerodynamic flow pattern around bluff bodies. However, their performance and optimal configurations need further investigations.

Combining natural ventilation with other passive cooling technologies, such as thermal mass [176,177], phase change materials [178], solar chimney [179,180], double-skin wall [181,182] and earth-air-tube ventilation [183] attracted more and more attentions in recent years. These advanced passive cooling systems in combination with standby mechanical ventilation systems would definitely improve ventilation rate, reduce pollutant and noise penetration, and be less affected by climate change and more frequent occurrence of extremely climatic events [184]. However, application of these technologies in urban buildings particularly in densely populated urban areas is limited by many factors, such as natural resources availability, integration with buildings, wind safety, cost, operation and maintenance.

4.2. Indoor air quality

Existing studies (see Sections 2.3 and 3.2) have revealed that (a) the concentration levels of traffic-related pollutants in street canyons are mostly 1.0-3.0 times higher than urban background concentrations and (b) the fractions of outdoor pollutants entering indoors are dozens of percentages. These results indicate explicitly the importance of taking the outdoor air quality into account when designing buildings with open windows or doors connecting indoor and outdoor. Given that the high pollutant concentration accumulated inside street canyons due to increased traffic-related sources and adverse dispersion condition, the penetration of outdoor pollutants through open windows or doors into the indoor environment should be paid particular attentions. Based on current knowledge, the possible implications and suggestions can be summarized as follows.

First, it has been evident that pollutant dilution and dispersion in street canyons is strongly dependent on the flow exchange between the canyon interior and its above atmosphere. Again, urban planning should be carefully conducted to ensure that coupling between the canyon flow and its above ambient flow can be frequently established. In addition, buildings that frequently use natural ventilation, such as residential buildings, should not be planned in high traffic intensity areas. Building design should consider its surrounding traffic conditions; particularly flow intakes should not orient major traffic streets.

Second, similar with controlling indoor heat load to promote the use of natural ventilation, a reasonable control of indoor activities to decrease indoor pollutants load is very helpful for promoting the use of natural ventilation, as indoor activities are a main contributor of indoor pollutants.

Third, although existing studies provide sufficient qualitative evidences regarding the penetration of outdoor pollutants into the indoor environment, quantitative studies are still very few. More quantitative studies should be conducted to clarify the penetration quantity and its influencing factors. Particularly, the measured indoor traffic-related pollutants should be compared with local IAQ standards.

Fourth, similarly with the third point, the dynamic process of outdoor pollutants penetrating into the indoor environment is far from well understood. Here, particulate and gaseous pollutants should be distinguished. The main objectives of understanding the dynamic penetration process are to clarify the penetration mechanisms and the influencing factors of penetration, which are the fundamental information for formulating effective control measures. In addition to traffic pollutant sources, which are almost volumetric sources, some point sources, such as exhausts of parking garages, restaurant kitchens and laundry shops, as well as even intentional releases by terrorists, are important outdoor pollutants that could have a local effect on their nearby indoor environments. More studies on these aspects are necessary.

Fifth, mechanical ventilation with filters can significantly reduce the penetration of outdoor particulate pollutants. Park et al. [18] reported that, compared with use of natural ventilation, use of mechanical ventilation with filters in residential buildings reduce I/ O ratios by 26% for submicron particles and 65% for fine particles, which substantially reduces the exposure to outdoor particles. The rationale of a hybrid ventilation system is to allow maximum use of natural ventilation, while mechanical ventilation system works only when natural ventilation cannot provide the required IEQ. Good performance of hybrid ventilation with filters and tempers in residential buildings was reported recently by Turner and Awbi [169]. In general, hybrid ventilation system is a simple, feasible, low energy solution for building design, which (a) ensures sufficient ventilation rates regardless of the magnitude of outdoor wind speeds and indoor/outdoor temperature differences, (b) reduces significantly exposure to particles when the outdoor particulate pollutants' concentration levels are unacceptable for natural ventilation, and (c) can be used in almost all situations where natural ventilation can be used. Further development on hybrid ventilation should focus on its standardized application, particularly with automatic control, in residential and school buildings.

4.3. Indoor traffic noise

Existing knowledge (see Section 2.4) has indicated that average noise levels due to road traffic in the lower part of street canyons easily exceed 70 dBA, which would certainly result in unfavorable indoor sound environment in nearby naturally ventilated buildings. In this context, more studies should be conducted to increase the understanding of sound environment in street canyons, particularly the vertical variation of noise level and its relationship with influencing factors. In addition, noise control of naturally ventilated buildings, particularly ventilation openings, is worthy of further investigations [185]. It is evident that the most efficient method to attenuate noise is to locate the building or the windows in an area of low noise levels. Within a high noise area, typical noise control methods at an window is to present barriers, such as screen, balcony and acoustic louvre [185], and to incorporate reactive elements [186], absorptive materials [187] and ceiling-mounted reflectors [188]. These measures can generally lead to a noise reduction of up to 23 dBA [187]. Moreover, Azkorra et al. [189] found that green walls have a weighted sound reduction potential of 15 dBA.

Acoustic performance of an opening system should be provided, together with ventilation performance, to designers so that they could select an approach to satisfy both. Oldham et al. [190] proposed a technique to enable the design of acoustic and ventilation performance of opening systems simultaneously. However, there is still very few information available on the combined analysis of noise attenuation and flow characteristics of a control method. It is to date still difficult to ensure an acoustically comfortable indoor environment and an adequate flow exchange rate simultaneously in a naturally ventilated building in a high traffic intensity area.

5. Summary and conclusions

This paper provides a review of past studies on (a) the

microclimate in urban street canyons, (b) the potential influence of such microclimate on IEQ in the nearby naturally ventilated buildings, and (c) the real-life IEQ statuses in naturally ventilated urban buildings. Based on existing knowledge, implications for urban planning and building design are summarized and recommendations for future studies are made.

In street canyons with AR greater than 0.65-0.7, wind directions are mostly parallel (or closely parallel) to building facades in both vertical and horizontal directions; wind speeds are mostly less than 1.0 m/s, decreased up to approximately 70-90% compared to those above the street canyons. Dependent on meteorological condition and canyon characteristics, the recorded simultaneous temperature difference between two facades reach up to $19 \,^\circ$ C, while air temperature difference inside a canyon is very slight. Street canyons are hotspots for both traffic-related pollutants and noise. Mean pollutant concentrations in street canyons are mostly 1.0-3.0 times higher than corresponding urban background concentrations, while mean noise level in the lower part of a canyon easily exceeds the threshold of 70 dBA.

The ACH value of naturally ventilated buildings is significantly decreased (up to nearly 100%) in urban areas compared to that in isolated conditions. Natural ventilation sometimes fails to maintain sufficient ACH values for acceptable IAQ in urban buildings, although it can potentially provide very high (dozens of) ACH values. Traffic-related pollutants have, up to 100%, higher concentrations in high traffic intensity buildings than in low traffic intensity buildings, where the penetration of outdoor pollutants increases the indoor pollutant concentrations by dozens of percentages.

Existing knowledge suggests that reasonable urban planning and careful envelope design are the primary ways to improve the IEQ in naturally ventilated buildings and thus maximize the use of natural ventilation. For pollutant and noise, naturally ventilated buildings should be planned in low traffic intensity areas and the main windows should not orient heavy traffic streets. Hybrid ventilation using mechanical ventilation with filtrations as a standby system for natural ventilation is a promising ventilation system in urban areas to ensure a minimum ventilation rate and avoid excessive penetration of outdoor pollutants.

However, quantitative studies, particularly on-site measurements, of both the microclimate in street canyons and the IEQ in naturally ventilated urban buildings are still very few, and most of them are short-term measurements under a specific environmental condition. Cross comparison of these studies and analysis of influencing factors are currently impossible. In order to better understand and improve IEQ in naturally ventilated urban buildings, the following recommendations are made for future studies.

- Large-scale, long-term, high-resolution, standardized on-site measurements in typical urban street canyons should be conducted to accumulate microclimatic data, which are fundamental information for urban planning, building design, and even CFD simulations.
- More on-site measurements should also be conducted to increase the understanding of IEQ in naturally ventilated urban buildings. Measured data should be compared to local IEQ standards and should be analyzed to summarize influencing factors.
- Dynamic penetration process of particulate and gaseous pollutants from the street canyons to the nearby indoor environment should be investigated to understand the penetration mechanisms and generate design guidelines.
- Combined analysis of flow characteristics, pollutant penetration, and noise attenuation should be conducted to optimize building envelope design to adapt street canyon microclimate.

• More efforts should be made to solve related issues of combining natural ventilation with other active and/or passive cooling technologies in urban buildings.

Acknowledgment

The review work described in this paper was conducted when the first author, Dr Zhengtao Ai, was a Postdoctoral Fellow in The Hong Kong Polytechnic University, who was financially supported by a grant from the Environment and Conservation Fund of Hong Kong (Project ECF 18/2014).

References

- F. Allard, M. Santamouris, Natural Ventilation in Buildings: a Design Handbook, James and James Science Publishers, London, 1998.
- [2] B.R. Hughes, H.N. Chaudhry, S.A. Ghani, A review of sustainable cooling technologies in buildings, Renew. Sustain. Energy Rev. 15 (2011) 3112–3120.
- [3] D. Etheridge, A perspective on fifty years of natural ventilation research, Build. Environ. 91 (2015) 51–60.
- [4] BRECSU, Energy Consumption Guide 19: Energy Use in Offices, British Research Establishment Conservation Support Unit, Garston, Watford, UK, 2000.
- [5] I. Oropeza-Perez, P.A. Ostergaard, Potential of natural ventilation in temperate countries – A case study of Denmark, Appl. Energy 114 (2014) 520–530.
- [6] I. Oropeza-Perez, P.A. Ostergaard, Energy saving potential of utilizing natural ventilation under warm conditions – A case study of Mexico, Appl. Energy 130 (2014) 20–32.
- [7] A. Aflaki, N. Mahyuddin, Z.A.C. Mahmoud, M.R. Baharum, A review on natural ventilation applications through building facade components and ventilation openings in tropical climates, Energy Build. 101 (2015) 153–162.
- [8] C. Simonson, Energy consumption and ventilation performance of a naturally ventilated ecological house in a cold climate, Energy Build. 37 (2005) 23–35.
- [9] J.J. Jaakkola, P. Miettinen, Type of ventilation system in office buildings and sick building syndrome, Am. J. Epidemiol. 141 (8) (1995) 755–765.
- [10] O. Seppanen, W.J. Fisk, Association of ventilation system type with SBS symptoms in office workers, Indoor Air 12 (2002) 98–112.
- [11] C. Ghiaus, F. Allard, Natural Ventilation in the Urban Environment: Assessment and Design, James and James Science Publishers, London, 2005.
- [12] P. Wargocki, D.P. Wyon, J. Sundell, G. Clausen, P.O. Fanger, The effects of outdoor air supply rate in an office on perceived air quality, sick building syndrome (SBS) symptoms and productivity, Indoor Air 10 (2000) 222–236.
- [13] A.R. Escombe, C.C. Oeser, R.H. Gilman, M. Navincopa, E. Ticona, W. Pan, et al., Natural ventilation for the prevention of airborne contagion, PLoS Med. 4 (2) (2007) e68.
- [14] Y. Li, G.M. Leung, J.W. Tang, X. Yang, C.Y.H. Chao, J.Z. Lin, et al., Role of ventilation in airborne transmission of infectious agents in the built environment – a multidisciplinary systematic review, Indoor Air 17 (2007) 2–18.
- [15] J. Sundell, H. Levin, W.W. Nazaroff, W.S. Cain, W.J. Fisk, D.T. Grimsrud, et al., Ventilation rates and health: multidisciplinary review of the scientific literature, Indoor Air 21 (2011) 191–204.
- [16] Z. Lin, S. Deng, The outdoor air ventilation rate in high-rise residences employing room air conditioners, Build. Environ. 38 (2003) 1389–1399.
- [17] H. Li, X. Li, M. Qi, Field testing of natural ventilation in college student dormitories (Beijing, China), Build. Environ. 78 (2014) 36–43.
- [18] J.S. Park, N.Y. Jee, J.W. Jeong, Effects of types of ventilation system on indoor particle concentrations in residential buildings, Indoor Air 24 (2014) 629–638.
- [19] N.T. Chao, W.A. Wang, C.M. Chiang, A study of a control strategy utilizing outdoor air to reduce the wintertime carbon dioxide levels in a typical Taiwanese bedroom, Energy Build. 29 (1998) 93–105.
- [20] M. Santamouris, D.N. Asimakopoulos, Passive Cooling of Buildings, James and James Science Publishers, London, 1996.
- [21] M. Liddament, The applicability of natural ventilation Technical Editorial, Int. J. Vent. 8 (3) (2009) 189–199.
- [22] C. Ghiaus, F. Allard, M. Santamouris, C. Georgakis, F. Nicol, Urban environment influence on natural ventilation potential, Build. Environ. 41 (2006) 395–406.
- [23] T.R. Oke, Boundary Layer Climates, second ed., Routledge, New York, 1987.
- [24] HKPD, Feasibility Study for Establishment of Air Ventilation Assessment System, Final Report, the Government of the Hong Kong Special Administrative Region, Hong Kong Planning Department, 2005. http://www.pland. gov.hk/pland_en/p_study/comp_s/avas/papers%26reports/final_report.pdf. page 96.
- [25] C. Georgakis, M. Santamouris, Experimental investigation of air flow and temperature distribution in deep urban canyons for natural ventilation purposes, Energy Build. 38 (2006) 367–376.
- [26] E. Andreou, K. Axarli, Investigation of urban canyon microclimate in traditional and contemporary environment: experimental investigation and parametric analysis, Renew. Energy 43 (2012) 354–363.

- [27] S. Weber, W. Kuttler, K. Weber, Flow characteristics and particle mass and number concentration variability within a busy urban street canyon, Atmos. Environ. 40 (2006) 7565–7578.
- [28] S.H.L. Yim, J.C.H. Fung, A.K.H. Lau, S.C. Kot, Air ventilation impacts of the "wall effect" resulting from the alignment of high-rise buildings, Atmos. Environ. 43 (2009) 4982–4994.
- [29] J. Spadaro, A. Rabl, Damage costs due to automotive air pollution and the influence of street canyons, Atmos. Environ. 35 (2001) 4763-4775.
- [30] R. van Dingenen, F. Raes, J.P. Putaud, U. Baltensperger, A. Charron, M.C. Facchini, et al., A European aerosol phenomenology – 1: physical characteristics of particulate matter at Kerbside, urban, rural and background sites in Europe, Atmos. Environ. 38 (16) (2004) 2561–2577.
- [31] H. Boogaard, N.A.H. Janssen, P.H. Fischer, G.P.A. Kos, E.P. Weijers, F.R. Cassee, et al., Contrasts in oxidative potential and other particulate matter characteristics collected near major streets and background locations, Environ. Health Perspect. 120 (2) (2012) 185–191.
- [32] C. Johansson, M. Norman, L. Burman, Road traffic emission factors for heavy metals, Atmos. Environ. 43 (2009) 4681–4688.
- [33] J.K. Gietl, R. Lawrence, A.J. Thorpe, R.M. Harrison, Identification of brake wear particles and derivation of a quantitative tracer for brake dust at a major road, Atmos. Environ. 44 (2010) 141–146.
- [34] EPA, EPA Identifies Noise Levels Affecting Health and Welfare, United States Environmental Protection Agency, 2015. http://www2.epa.gov/aboutepa/ epa-identifies-noise-levels-affecting-health-and-welfare (accessed on August 01).
- [35] Z. Ross, I. Kheirbek, J.E. Clougherty, K. Ito, T. Matte, S. Markowitz, et al., Noise, air pollutants and traffic: continuous measurement and correlation at a hightraffic location in New York City, Environ. Res. 111 (2011) 1054–1063.
- [36] T.P. McAlexander, R.R.M. Gershon, R.L. Neitzel, Street-level noise in an urban setting: assessment and contribution to personal exposure, Environ. Health 14 (2015) 18.
- [37] W.H. Snyder, Guideline for Fluid Modeling of Atmospheric Diffusion, Meteorology and Assessment, Research Triangle Park, NC, Division Environmental Sciences Research Laboratory, U.S. Environmental Protection Agency, 1981.
- [38] B. Blocken, T. Stathopoulos, J. Carmeliet, J. Hensen, Application of CFD in building performance simulation for the outdoor environment: an overview, J. Build. Perform. Simul. 4 (2) (2011) 157–184.
- [39] B. Blocken, 50 years of computational wind engineering: past, present and future, J. Wind Eng. Ind. Aerodyn. 129 (2014) 69–102.
- [40] M. Santamouris, Energy and Climate in the Urban Environment, James and James Science Publishers, London, 2001.
- [41] Y. Nakamura, T.R. Oke, Wind, temperature and stability conditions in an eastwest oriented urban canyon, Atmos. Environ. 22 (12) (1988) 2691–2700.
- [42] A.J. Arnfield, G. Mills, An analysis of the circulation characteristics and energy budget of a dry, asymmetric, east, west urban canyon, 1 Circulation characteristics, Int. J. Climatol. 14 (1994) 119–134.
- [43] I.D. Longley, M.W. Gallagher, J.R. Dorsey, M. Flynn, J.F. Barlow, Short-term measurements of airflow and turbulence in two street canyons in Manchester, Atmos. Environ. 38 (2004) 69–79.
- [44] C. Georgakis, M. Santamouris, On the airflow in urban canyons for ventilation purposes, Int. J. Vent. 3 (1) (2004) 53–65.
- [45] I. Eliasson, B. Offerle, C.S.B. Grimmond, S. Lindqvist, Wind fields and turbulence statistics in an urban street canyon, Atmos. Environ. 40 (2006) 1–16.
- [46] F.T. DePaul, C.M. Sheih, Measurements of wind velocities in a street canyon, Atmos. Environ. 20 (3) (1986) 455–459.
- [47] R.A. Mc Cormick, Air pollution in the locality of buildings, Philos. Trans. R. Soc. Lond. A 269 (1971) 515–526.
- [48] R.J. Yamartino, G. Wiegand, Development and evaluation of simple models for the flow, turbulence and pollution concentration fields within an urban street canyon, Atmos. Environ. 20 (1986) 2137–2156.
- [49] M. Santamouris, N. Papanikolaou, I. Koronakis, I. Livada, D. Asimakopoulos, Thermal and air flow characteristics in a deep pedestrian canyon under hot weather conditions, Atmos. Environ. 33 (1999) 4503–4521.
- [50] C. Georgakis, M. Santamouris, On the estimation of wind speed in urban canyons for ventilation purposes – Part 1: Coupling between the undisturbed wind speed and the canyon wind, Build. Environ. 43 (2008) 1404–1410.
- [51] M. Santamouris, C. Georgakis, A. Niachou, On the estimation of wind speed in urban canyons for ventilation purposes – Part 2: using of data driven techniques to calculate the more probable wind speed in urban canyons for low ambient wind speeds, Build. Environ. 43 (2008) 1411–1418.
- [52] M. Pidwirny, Köppen Climate Classification System, 2011. Retrieved from, http://www.eoearth.org/view/article/162263 (accessed 14.10.15).
- [53] A.J. Manning, K.J. Nicholson, D.R. Middleton, S.C. Rafferty, Field study of wind and traffic to test a street canyon pollution model, Environ. Monit. Assess. 60 (2000) 283–313.
- [54] P.S.J. Buller, Wind Speeds Measured within an Urban Area, Report No. 2, Building Establishment Note N12/76, DoE, Building Research Station, Watford, UK, 1976.
- [55] K. Niachou, I. Livada, M. Santamouris, Experimental study of temperature and airflow distribution inside an urban street canyon during hot summer weather conditions – Part 2: airflow analysis, Build. Environ. 43 (2008) 1393–1403.
- [56] S. Kitous, R. Bensalem, L. Adolphe, Airflow patterns within a complex urban topography under hot and dry climate in the Algerian Sahara, Build. Environ.

56 (2012) 162-175.

- [57] T. Kubota, M. Miura, Y. Tominaga, A. Mochida, Wind tunnel tests on the relationship between building density and pedestrian-level wind velocity: development of guidelines for realizing acceptable wind environment in residential neighborhoods, Build. Environ. 43 (2008) 1699–1708.
- [58] K. Niachou, I. Livada, M. Santamouris, Experimental study of temperature and airflow distribution inside an urban street canyon during hot summer weather conditions – Part 1: air and surface temperatures, Build. Environ. 43 (2008) 1383–1392.
- [59] D. Pearlmutter, A. Bitan, P. Berliner, Microclimatic analysis of 'compact' urban canyons in an arid zone, Atmos. Environ. 33 (1999) 4143–4150.
- [60] F. Bourbia, H.B. Awbi, Building cluster and shading in urban canyon for hot dry climate Part 1: air and surface temperature measurements, Renew. Energy 29 (2004) 249–262.
- [61] I. Eliasson, Urban geometry, surface temperature and air temperature, Energy Build. 15–16 (1990) 141–145.
- [62] I. Eliasson, Urban nocturnal temperatures, street geometry and land use, Atmos. Environ. B 30 (3) (1996) 379–392.
- [63] G. Mills, A.J. Arnfield, Simulation of energy budget of urban canyon, comparison of model results with measurements, Atmos. Environ. 27B (1993) 171–181.
- [64] F. Bourbia, H.B. Awbi, Building cluster and shading in urban canyon for hot dry climate Part 2: shading simulations, Renew. Energy 29 (2004) 291–301.
- [65] B. Givoni, Climate Considerations in Building and Urban Design, Johan Wiley and Sons, New York, 1998.
- [66] L. Barring, J.O. Mattson, S. Lindovist, Canyon geometry, street temperatures and urban heat island in Malmo Sweden, Int. J. Climatol. 5 (1985) 433–444.
 [67] L. Shashua-Bar, M.E. Hoffman, Geometry and orientation aspects in passive
- cooling of canyon streets with trees, Energy Build. 35 (2003) 61–68.
- [68] G. Mihalakakou, H. Flocas, M. Santamouris, C. Helmis, The impact of synoptic scale atmospheric circulation on the urban heat island effect over Athens, Greece, J. Appl. Meteorol. 41 (5) (2002) 519–527.
- [69] G. Mihalakakou, M. Santamouris, N. Papanikolaou, C. Cartalis, Simulation of the urban heat island phenomenon in Mediterranean climates, Pure Appl. Geophys. 161 (2004) 429–451.
- [70] H. Akbari, A. Rosenfeld, H. Taha, Recent developments in heat island studies: technical and policy, in: Proceedings of the Workshop on Saving Energy and Reducing Atmospheric Pollution by Controlling Summer Heat Islands, Berkeley, CA, February 23-24, 1989.
- [71] M. Bojic, M. Lee, F. Yik, Flow and temperatures outside a high-rise residential building due to heat rejection by its air-conditioners, Energy Build. 33 (2001) 737–751.
- [72] H.U. Pfeffer, Ambient air concentrations of pollutants at traffic-related sites in urban areas of North Rhine-West-phalia, Germany, Sci. Total Environ. 146/ 147 (1994) 263–273.
- [73] A.G. Clark, Y.H. Ko, The relative significance of vehicular emissions and other emissions of volatile organic compounds in the urban area of Leeds, UK, Sci. Total Environ. 189/190 (1996) 401–407.
- [74] Y.H. Zhang, S.D. Xie, L.M. Zeng, H.X. Wang, K.H. Yu, X.J. Zhu, et al., Traffic emission and its impact on air quality in Guangzhou area, J. Environ. Sci. 11 (1999) 355–360.
- [75] R.N. Colvile, E.J. Hutchinson, J.S. Mindell, R.F. Warren, The transport sector as a source of air pollution, Atmos. Environ. 35 (2001) 1537–1565.
- [76] T. Wang, S. Xie, Assessment of traffic-related air pollution in the urban streets before and during the 2008 Beijing Olympic Games traffic control period, Atmos. Environ. 43 (2009) 5682–5690.
- [77] S. Vardoulakis, B.E.A. Fisher, K. Pericleous, N. Gonzalez-Flesca, Modelling air quality in street canyons: a review, Atmos. Environ. 37 (2003) 155–182.
- [78] A. Thorpe, R.M. Harrison, Sources and properties of non-exhaust particulate matter from road traffic: a review, Sci. Total Environ. 400 (2008) 270–282.
- [79] D.B. Kittelson, W.F. Watts, J.P. Johnson, Nanoparticle emissions on Minnesota highways, Atmos. Environ. 38 (1) (2004) 9–19.
- [80] D.B. Kittelson, W.F. Watts, J.P. Johnson, M.L. Remerowki, E.E. Ische, G. Oberdorster, et al., On-road exposure to highway aerosols: 1 Aerosol and gas measurements, Inhal. Toxicol. 16 (S1) (2004) 31–39.
- [81] N. Bukowiecki, J. Dommen, A.S.H. Prevot, R. Richter, E. Weingartner, U. Baltensperger, A mobile pollutant measurement laboratory – measuring gas phase and aerosol ambient concentrations with high spatial and temporal resolution, Atmos. Environ. 36 (2002) 5569–5579.
- [82] E.P. Weijers, A.Y. Khlystov, G.P.A. Kos, J.W. Érisman, Variability of particulate matter concentrations along roads and motorways determined by a moving measurement unit, Atmos. Environ. 38 (2004) 2993–3002.
- [83] L. Pirjola, T. Parviainen, T. Hussein, A. Valli, K. Hameri, P. Aaalto, et al., "Sniffer" – a novel tool for chasing vehicles and measuring traffic pollutants, Atmos. Environ. 38 (2004) 3625–3635.
- [84] L.M. Zwack, C.J. Paciorek, J.D. Spengler, J.I. Levy, Characterizing local traffic contributions to particulate air pollution in street canyons using mobile monitoring techniques, Atmos. Environ. 45 (15) (2011) 2507–2514.
- [85] L.M. Zwack, C.J. Paciorek, J.D. Spengler, J.I. Levy, Modeling spatial patterns of traffic-related air pollutants in complex urban terrain, Environ. Health Perspect. 119 (6) (2011) 852–859.
- [86] L. Zhao, X. Wang, Q. He, H. Wang, G. Sheng, L. Chan, et al., Exposure to hazardous volatile organic compounds, PM₁₀ and CO while walking along streets in urban Guangzhou, China, Atmos. Environ. 38 (2004) 6177–6184.
- [87] Z. Qian, J. Zhang, F. Wei, W.E. Wilson, R.S. Chapman, Long-term ambient air

pollution levels in four Chinese cities: inter-city and intra-city concentration gradients for epidemiological studies, J. Expo. Anal. Environ. Epidemiol. 11 (2001) 341–351.

- [88] K.A. Kourtidis, I. Ziomas, C. Zerefos, E. Kosmidis, P. Symeonidis, E. Christophilopoulos, et al., Benzene, toluene, ozone, NO₂ and SO₂ measurements in an urban street canyon in Thessaloniki, Greece, Atmos. Environ. 36 (2002) 5355–5364.
- [89] G. Hoek, B. Brunekreef, A. Verhoeff, J. van Wijnen, P. Fischer, Daily mortality and air pollution in the Netherlands, J. Air Waste Manag. Assoc. 50 (2000) 1380–1389.
- [90] F. Nyberg, P. Gustavsson, L. Jarup, T. Bellander, N. Berglind, R. Jakobsson, et al., Urban air pollution and lung cancer in Stockholm, Epidemiology 11 (5) (2000) 487–495.
- [91] L.J. Hunter, G.T. Johnson, I.D. Watson, An investigation of three-dimensional characteristics of flow regimes within the urban canyon, Atmos. Environ. 26B (4) (1992) 425–432.
- [92] C.H. Liu, D.Y.C. Leung, M.C. Barth, On the prediction of air and pollutant exchange rates in street canyons of different aspect ratios using large-eddy simulation, Atmos. Environ. 39 (9) (2005) 1567–1574.
- [93] C.M.N. Riain, B. Fisher, C.J. Martin, J. Littler, Flow field and pollution dispersion in a central London street, Environ. Monit. Assess. 52 (1998) 299-314.
- [94] J.J. Kim, J.J. Baik, A numerical study of the effects of ambient wind direction on flow and dispersion in urban street canyons using the RNG k-e turbulence model, Atmos. Environ. 38 (19) (2004) 3039–3048.
- [95] Y. Qin, S.C. Kot, Dispersion of vehicular emission in street canyons, Guangzhou city, South China (P.R.C, Atmos. Environ. 27B (1993) 283–291.
- [96] E. Vignati, R. Berkowicz, O. Hertel, Comparison of air quality in streets of Copenhagen and Milan, in view of the climatological conditions, Sci. Total Environ. 189/190 (1996) 467–473.
- [97] S.G. Jones, B.E.A. Fisher, N. Gonzalez-Flesca, R. Sokhi, The use of measurement programmes and models to assess concentrations next to major roads in urban areas, Environ. Monit. Assess. 64 (2000) 531–547.
- [98] S. Kaur, M. Nieuwenhuijsen, R. Colvile, Fine particulate matter and carbon monoxide exposure concentrations in urban street transport microenvironments, Atmos. Environ. 41 (2007) 4781–4810.
- [99] D. Westerdahl, S. Fruin, T. Sax, P.M. Fine, C. Sioutas, Mobile platform measurements of ultrafine particles and associated pollutant concentrations on freeways and residential streets in Los Angeles, Atmos. Environ. 39 (2005) 3597–3610.
- [100] M.Y. Tsai, K.S. Chen, Measurements and three-dimensional modeling of air pollutant dispersion in an urban street canyon, Atmos. Environ. 38 (2004) 5911–5924.
- [101] A. Rakowska, K.C. Wong, T. Townsend, K.L. Chan, D. Westerdahl, S. Ng, et al., Impact of traffic volume and composition on the air quality and pedestrian exposure in urban street canyon, Atmos. Environ. 98 (2014) 260–270.
- [102] Y. Cheng, S.C. Lee, Y. Gao, L. Cui, W. Deng, J. Cao, et al., Real-time measurements of PM_{2.5}, PM_{10-2.5}, and BC in an urban street canyon, Particuology 20 (2015) 134–140.
- [103] X.M. Wang, G.Y. Sheng, J.M. Fu, C.Y. Chan, S.C. Lee, L.Y. Chan, et al., Urban roadside aromatic hydrocarbons in three cities of the Pearl River Delta, People's Republic of China, Atmos. Environ. 36 (2002) 5141–5148.
- [104] L.Y. Chan, W.L. Lau, X.M. Wang, J.H. Tang, Preliminary measurements of aromatic VOCs in public transportation modes in Guangzhou, China, Environ. Int. 29 (2003) 429–435.
- [105] F.T. DePaul, C.M. Sheih, A tracer study of dispersion in an urban street canyon, Atmos. Environ. 19 (4) (1985) 555–559.
- [106] S. Vardoulakis, N. Gonzalez-Flesca, B.E.A. Fisher, Assessment of traffic-related air pollution in two street canyons in Paris: implications for exposure studies, Atmos. Environ. 36 (2002) 1025–1039.
- [107] M. Vakeva, K. Hameri, M. Kulmala, R. Lahdes, J. Ruuskanen, T. Laitinen, Street level versus rooftop concentrations of submicron aerosol particles and gaseous pollutants in an urban street canyon, Atmos. Environ. 33 (1999) 1385–1397.
- [108] L.Y. Chan, W.S. Kwok, Vertical dispersion of suspended particulates in urban area of Hong Kong, Atmos. Environ. 34 (2000) 4403–4412.
- [109] H. Boogaard, G.P.A. Kos, E.P. Weijers, N.A.H. Janssen, P.H. Fischer, S.C. van der Zee, et al., Contrast in air pollution components between major streets and background locations: particulate matter mass, black carbon, elemental composition, nitrogen oxide and ultrafine particle number, Atmos. Environ. 45 (2011) 650–658.
- [110] T. Nielsen, Traffic contribution of polycyclic aromatic hydrocarbons in the center of a large city, Atmos. Environ. 20 (1996) 3481–3490.
- [111] C.C. Chan, H. Ozkaynak, J.D. Spengler, L. Sheldon, Driver exposure to volatile organic compounds, CO, Ozone, and NO₂ under different driving conditions, Environ. Sci. Technol. 25 (1991) 964–972.
- [112] M.S. Hammer, T.K. Swinburn, R.L. Neitzel, Environmental noise pollution in the United States: developing an effective public health response, Environ. Health Perspect. 122 (2) (2014) 115–119.
- [113] T. Bodin, M. Albin, J. Ardo, E. Stroh, P.O. Ostergren, J. Bjork, Road traffic noise and hypertension: results from a cross-sectional public health survey in southern Sweden, Environ. Health 8 (2009) 38.
- [114] G.A. Belojevic, B.D. Jakovljevic, V.J. Stojanov, V.Z. Slepcevic, K.Z. Paunovic, Nighttime road-traffic noise and arterial hypertension in an urban population, Hypertens. Res. 31 (4) (2008) 775–781.
- [115] B. Jakovljevic, G. Belojevic, K. Paunovic, V. Stojanov, Road traffic noise and

sleep disturbances in an urban population: cross-sectional study, Croat. Med. J. 47 (2006) 125–133.

- [116] X. Morelli, M. Foraster, I. Aguilera, X. Basagana, E. Corradi, A. Deltell, et al., Short-term associations between traffic-related noise, particle number and traffic flow in three European cities, Atmos. Environ. 103 (2015) 25–33.
- [117] F. Nicol, M. Wilson, The effect of street dimensions and traffic density on the noise level and natural ventilation potential in urban canyons, Energy Build. 36 (2004) 423–434.
- [118] C.M. Mak, W.K. Leung, G.S. Jiang, Measurement and prediction of road traffic noise at different building floor levels in Hong Kong, Build. Serv. Eng. Res. Technol. 31 (2) (2010) 131–139.
- [119] E. Gaja, A. Gimenez, S. Sancho, A. Reig, Sampling techniques for the estimation of the annual equivalent noise level under urban traffic conditions, Appl. Acoust. 64 (2003) 43–53.
- [120] A. Calixto, F.B. Diniz, P.H.T. Zannin, The statistical modeling of road traffic noise in an urban setting, Cities 20 (1) (2003) 23–29.
- [121] H. Qian, Y. Li, W.H. Seto, P. Ching, W.H. Chiang, H.Q. Sun, Natural ventilation for reducing airborne infection in hospitals, Build. Environ. 45 (2010) 559–565.
- [122] T. van Hooff, B. Blocken, On the effect of wind direction and urban surroundings on natural ventilation of a large semi-enclosed stadium, Comput. Fluids 39 (2010) 1146–1155.
- [123] C.F. Gao, W.L. Lee, The influence of surrounding buildings on the natural ventilation performance of residential dwellings in Hong Kong, Int. J. Vent. 11 (3) (2012) 297–310.
- [124] R. Ramponi, I. Gaetani, A. Angelotti, Influence of the urban environment on the effectiveness of natural night-ventilation of an office building, Energy Build. 78 (2014) 25–34.
- [125] M. Bady, S. Kato, T. Takahashi, H. Huang, Experimental investigations of the indoor natural ventilation for different building configurations and incidences, Build. Environ. 46 (2011) 65–74.
- [126] J.O.P. Cheung, C.H. Liu, CFD simulations of natural ventilation behavior in high-rise buildings in regular and staggered arrangements at various spacings, Energy Build. 43 (2011) 1149–1158.
- [127] K. Niachou, S. Hassid, M. Santamouris, I. Livada, Comparative monitoring of natural, hybrid and mechanical ventilation systems in urban canyons, Energy Build. 37 (2005) 503–513.
- [128] K. Niachou, S. Hassid, M. Santamouris, I. Livada, Experimental performance investigation of natural, mechanical and hybrid ventilation in urban environment, Build. Environ. 43 (2008) 1373–1382.
- [129] Z.T. Ai, C.M. Mak, D.J. Cui, On-site measurements of ventilation performance and indoor air quality in naturally ventilated high-rise residential buildings in Hong Kong, Indoor Built Environ. 24 (2) (2015) 214–224.
- [130] C.A. Gilkeson, M.A. Camargo-Valero, L.E. Pickin, C.J. Noakes, Measurement of ventilation and airborne infection risk in large naturally ventilated hospital wards, Build. Environ. 65 (2013) 35–48.
- [131] ASHRAE Standard 62.1, Ventilation for Acceptable Indoor Air Quality, American Society of Heating, Refrigerating and Air-conditioning Engineers, Inc, Atlanta, GA, 2013.
- [132] M. Santamouris, A. Synnefa, M. Asssimakopoulos, I. Livada, K. Pavlou, M. Papaglastra, et al., Experimental investigation of the air flow and indoor carbon dioxide concentration in classrooms with intermittent natural ventilation, Energy Build. 40 (2008) 1833–1843.
- [133] Z.T. Ai, C.M. Mak, J.L. Niu, Z.R. Li, The assessment of the performance of balconies using computational fluid dynamics, Build. Serv. Eng. Res. Technol. 32 (3) (2011) 229–243.
- [134] Z.T. Ai, C.M. Mak, J.L. Niu, Z.R. Li, Q. Zhou, The effect of balconies on ventilation performance of low-rise buildings, Indoor Built Environ. 20 (6) (2011) 649–660.
- [135] Z.T. Ai, C.M. Mak, J.L. Niu, Numerical investigation of wind-induced airflow and interunit dispersion characteristics in multistory residential buildings, Indoor Air 23 (2013) 417–429.
- [136] Z.T. Ai, C.M. Mak, A study of interunit dispersion around multistory buildings with single-sided ventilation under different wind directions, Atmos. Environ. 88 (2014) 1–13.
- [137] Z.T. Ai, C.M. Mak, Large eddy simulation of wind-induced interunit dispersion around multistory buildings, Indoor Air (2015), http://dx.doi.org/ 10.1111/ina.12200.
- [138] Z.T. Ai, C.M. Mak, Comparison of single-sided ventilation characteristics between single-storey and multi-storey buildings due to wind effect, Int. J. Vent. 14 (2) (2015) 181–190.
- [139] Z.R. Li, Z.T. Ai, W.J. Wang, Z.R. Xu, X.Z. Gao, H.S. Wang, Evaluation of airflow pattern in wind-driven naturally ventilated atrium buildings: measurement and simulation, Build. Serv. Eng. Res. Technol. 35 (5) (2014) 139–154.
- [140] Department of Health, Health Technical Memorandum HTM 03-01: Specialized Ventilation for Healthcare Premises, Part a: Design and Validation, The Stationary Office, London, 2007.
- [141] CDC, Guidelines for Environmental Infection Control in Health-care Facilities, Department of Health and Human Services Centers for Disease Control and Prevention (CDC), Atlanta, US, 2003.
- [142] C. Chen, B. Zhao, Review of relationship between indoor and outdoor particles: I/O ratio, infiltration factor and penetration factor, Atmos. Environ. 45 (2011) 275–288.
- [143] J. Wichmann, T. Lind, M.A.M. Nilsson, T. Bellander, PM_{2.5}, soot and NO₂ indoor-outdoor relationships at homes, pre-schools and schools in

Stockholm, Sweden, Atmos. Environ. 44 (2010) 4536-4544.

- [144] A. Challoner, L. Gill, Indoor/outdoor air pollution relationships in ten commercial buildings: PM_{2.5} and NO₂, Build. Environ. 80 (2014) 159–173.
 [145] C.Y. Chao, K.K. Wong, E.C. Cheng, Size distribution of indoor particulate
- [145] C.I. Chao, K.K. Wong, E.C. Cheng, Size distribution of index particular matter in 60 homes in Hong Kong, Indoor Built Environ. 11 (2002) 18–26.
 [146] B. Horemans, R.V. Grieken, Speciation and diurnal variation of thoracic, fine
- thoracic and sub-micrometer airborne particulate matter at naturally ventilated office environments, Atmos. Environ. 44 (2010) 1497–1505.
- [147] M. Habil, A. Taneja, Children's exposure to indoor particulate matter in naturally ventilated schools in India, Indoor Built Environ. 20 (4) (2011) 430–448.
- [148] Y. Zhu, L. Yang, C. Meng, Q. Yuan, C. Yan, C. Dong, et al., Indoor/outdoor relationships and diurnal/nocturnal variations in water-soluble ion and PAH concentrations in the atmospheric PM_{2.5} of a business office area in Jinan, a heavily polluted city in China, Atmos. Res. 153 (2015) 276–285.
- [149] H. Krasnov, I. Katra, V. Novack, A. Vodonos, M.D. Friger, Increased indoor PM concentrations controlled by atmospheric dust events and urban factors, Build. Environ. 87 (2015) 169–176.
- [150] Y. Han, M. Qi, Y. Chen, H. Shen, J. Liu, Y. Huang, et al., Influences of ambient air PM_{2.5} concentration and meteorological condition on the indoor PM_{2.5} concentrations in residential apartment in Beijing using a new approach, Environ. Pollut. 205 (2015) 307–314.
- [151] L. Yao, L. Yang, J. Chen, K. Toda, X. Wang, J. Zhang, et al., Levels, indooroutdoor relationships and exposure risks of airborne particle-associated perchlorate and chlorate in two urban areas in Eastern Asia, Chemosphere 135 (2015) 31–37.
- [152] P.H. Fischer, G. Hoek, H. van Reeuwijk, D.J. Briggs, E. Lebret, J.H. van Wijnen, et al., Traffic-related differences in outdoor and indoor concentrations of particles and volatile organic compounds in Amsterdam, Atmos. Environ. 34 (2000) 3713–3722.
- [153] S.Z. Sajani, I. Ricciardelli, A. Trentini, D. Bacco, C. Maccone, S. Castellazzi, et al., Spatial and indoor/outdoor gradients in urban concentrations of ultrafine particles and PM_{2.5} mass and chemical components, Atmos. Environ. 103 (2015) 307–320.
- [154] D. Lai, P. Karava, Q. Chen, Study of outdoor ozone penetration into buildings through ventilation and infiltration, Build. Environ. 93 (2015) 112–118.
- [155] I. Rivas, M. Viana, T. Moreno, M. Pandolfi, F. Amato, C. Reche, et al., Child exposure to indoor and outdoor air pollutants in schools in Barcelona, Spain, Environ. Int. 69 (2014) 200–212.
- [156] Y. Zhu, W.C. Hinds, M. Krudysz, T. Kuhn, J. Froines, C. Sioutas, Penetration of freeway ultrafine particles into indoor environments, Aerosol Sci. 36 (2005) 303–322.
- [157] L. Kliucininkas, E. Krugly, I. Stasiulaitiene, I. Radziuniene, T. Prasauskas, A. Jonusas, et al., Indoor-outdoor levels of size segregated particular matter and mono/polycyclic aromatic hydrocarbons among urban areas using solid fuels for heating, Atmos. Environ. 97 (2014) 83–93.
- [158] F. Amato, I. Rivas, M. Viana, T. Moreno, L. Bouso, C. Reche, et al., Sources of indoor and outdoor PM_{2.5} concentrations in primary schools, Sci. Total Environ. 490 (2014) 757–765.
- [159] R. Meier, C. Schindler, M. Eeftens, I. Aguilera, R.E. Ducret-Stich, A. Ineichen, et al., Modeling indoor air pollution of outdoor origin in homes of SAPALDIA subjects in Switzerland, Environ. Int. 82 (2015) 85–91.
- [160] J.M. Santos, I. Mavroidis, N.C. Reis Jr., E.C. Pagel, Experimental investigation of outdoor and indoor mean concentrations and concentration fluctuations of pollutants, Atmos. Environ. 45 (2011) 6534–6545.
- [161] F. Yang, Y. Kang, Y. Gao, K. Zhong, Numerical simulations of the effect of outdoor pollutants on indoor air quality of buildings next to a street canyon, Build. Environ. 87 (2015) 10–22.
- [162] W.R. Chang, C.L. Cheng, Carbon monoxide transport in an enclosed room with sources from a water heater in the adjacent balcony, Build. Environ. 43 (2008) 861–870.
- [163] RESHYVENT, Cluster Project on Demand Controlled Hybrid Ventilation in Residential Buildings with Specific Emphasis of the Integration of Renewables, Contract No. ENK6-CT2001-00533, European Joule Project, 2002.
- [164] Z.T. Ai, C.M. Mak, Modelling of coupled urban wind flow and indoor air flow on a high-density near-wall mesh: sensitivity analyses and case study for

single-sided ventilation, Environ. Modell. Softw. 60 (2014) 57-68.

- [165] Z.T. Ai, C.M. Mak, Analysis of fluctuating characteristics of wind-induced airflow through a single opening using LES modeling and the tracer gas technique, Build. Environ. 80 (2014) 249–258.
- [166] Y. Ji, K.J. Lomas, M.J. Cook, Hybrid ventilation for low energy building design in south China, Build. Environ. 44 (2009) 2245–2255.
- [167] R.Z. Homod, K.S.M. Sahari, Energy savings by smart utilization of mechanical and natural ventilation for hybrid residential building model in passive climate, Energy Build. 60 (2013) 310–329.
- [168] A. Dhalluin, K. Limam, Comparison of natural and hybrid ventilation strategies used in classrooms in terms of indoor environmental quality, comfort and energy savings, Indoor Built Environ. 23 (4) (2014) 527–542.
- [169] W.J.N. Turner, H.B. Awbi, Experimental investigation into the thermal performance of a residential hybrid ventilation system, Appl. Therm. Eng. 77 (2015) 142–152.
- [170] O. Saadatian, L.C. Haw, K. Sopian, M.Y. Sulaiman, Review of windcatcher technologies, Renew. Sust. Energy Rev. 16 (2012) 1477–1495.
- [171] B.R. Hughes, J.K. Calautit, S.A. Ghani, The development of commercial wind towers for natural ventilation: A review, Appl. Energy 92 (2012) 606–627.
- [172] M.F. Mohamed, The Potential of Balconies to Induce Wind-driven Natural Ventilation in Single-sided Ventilated High-rise Apartments, The University of the New South Wales, Australia, 2011 (PhD thesis).
- [173] N. Khan, Y. Su, S.B. Riffat, A review on wind driven ventilation techniques, Energy Build. 40 (2008) 1586–1604.
- [174] C.M. Mak, J.L. Niu, C.T. Lee, K.F. Chan, A numerical simulation of wing walls using computational fluid dynamics, Energy Build. 39 (2007) 995–1002.
- [175] P. Prajongsan, S. Sharples, Enhancing natural ventilation, thermal comfort and energy savings in high-rise residential buildings in Bangkok through the use of ventilation shafts, Build. Environ. 50 (2012) 104–113.
- [176] J. Zhou, G. Zhang, Y. Lin, Y. Li, Coupling of thermal mass and natural ventilation in buildings, Energy Build. 40 (2008) 979–986.
- [177] T. Yu, P. Heiselberg, B. Lei, M. Pomianowski, C. Zhang, A novel system solution for cooling and ventilation in office buildings: a review of applied technologies and a case study, Energy Build. 90 (2015) 142–155.
- [178] V.A.A. Raj, R. Velraj, Review on free cooling of buildings using phase change materials, Renew. Sust. Energ Rev. 14 (2010) 2819–2829.
- [179] D.J. Harris, N. Helwig, Solar chimney and building ventilation, Appl. Energy 84 (2007) 135–146.
- [180] X.Q. Zhai, Z.P. Song, R.Z. Wang, A review for the applications of solar chimneys in buildings, Renew. Sust. Energy Rev. 15 (2011) 3757–3767.
- [181] J. Zhou, Y. Chen, A review on applying ventilated double-skin facade to buildings in hot-summer and cold-winter zone in China, Renew. Sust. Energy Rev. 14 (2010) 1321–1328.
- [182] M.A. Shameri, M.A. Alghoul, K. Sopian, M.F.M. Zain, O. Elayeb, Perspectives of double skin facade systems in buildings and energy saving, Renew. Sust. Energy Rev. 15 (2011) 1468–1475.
- [183] D. Yang, J. Zhang, Theoretical assessment of the combined effects of building thermal mass and earth-air-tube ventilation on the indoor thermal environment, Energy Build. 81 (2014) 182–199.
- [184] M. Liddament, J. Axley, P. Heiselberg, Y. Li, T. Stathopoulos, Achieving natural and hybrid ventilation in practice, Int. J. Vent. 5 (1) (2006) 115–130.
- [185] C.D. Field, F.R. Fricke, The Attenuation of Noise Entering Buildings through Ventilation Openings, University of Sydney, Australia, 1997 (PhD thesis).
- [186] M.H.F. de Salis, D.J. Oldham, S. Sharples, Noise control strategies for naturally ventilated buildings, Build. Environ. 37 (2002) 471–484.
- [187] P.J. Lee, Y.H. Kim, J.Y. Jeon, K.D. Song, Effects of apartment building façade and balcony design on the reduction of exterior noise, Build. Environ. 42 (2007) 3517–3528.
- [188] T. Ishizuka, K. Fujiwara, Traffic noise reduction at balconies on a high-rise building facade, J. Acoust. Soc. Am. 131 (3) (2012) 2110–2117.
- [189] Z. Azkorra, G. Perez, J. Coma, L.F. Cabeza, S. Bures, J.E. Alvaro, et al., Evaluation of green walls as a passive acoustic insulation system for buildings, Appl. Acoust. 89 (2015) 46–56.
- [190] D.J. Oldham, M.H. de Salis, S. Sharples, Reducing the ingress of urban noise through natural ventilation openings, Indoor Air 14 (Suppl. 8) (2008) 118–126.